Author's response to reviews

Title: Oxidative and pre-inflammatory stress in wedge resection of pulmonary parenchyma using the radiofrequency ablation technique in a swine model.

Authors:

Theodoros Karaiskos (theokaraiskos@gmail.com)
Periklis Tomos (periklistomos@hotmail.com)
Irene Asouhidou (iasouhidou@aol.com)
Nikolaos Nikiteas (n Nikolaos.nikiteas@gmail.com)
Theodoros Kontakiotis (kontakiotis@gmail.com)
Apostolos Papalois (apo@hol.gr)

Version: 3 Date: 4 January 2012

Author's response to reviews:

I have made all the corrections asked by the editorial team. The author’s names and the affiliations have been changed according to the directions. Also it has been added the “competing interests” and the authors’ contribution section. All the other recommended corrections major and minor have been made.

Thank you

Sincerely

Dr.Irene Asouhidou